Permits

From: Julian Morse

Sent: Thursday, October 15, 2015 8:50 AM

To: Deborah Flemming

Cc: Tyree Mullaney; Jen Potten; Permits; Lindsey Cymbalisty; Angela Plautz

Subject: RE: NATCL - Comment Response Extension Request

Hi Deborah,

This is a reasonable request, the extension is granted.

Julian Morse

Regulatory Officer

Mackenzie Valley Land and Water Board

7th Floor, 4922 48th St. | PO Box 2130 | Yellowknife, NT | Canada | X1A 2P6

ph 867.766.7453 | mobile 867.446.0276 | fax 867.873.6610

jmorse@mvlwb.com | www.mvlwb.com

Please note: All correspondence to the Board, including emails, letters, faxes, and attachments are public documents and may be posted to the Public Registry.



From: Deborah Flemming [mailto:dflemming@natcl.ca]

Sent: Wednesday, October 14, 2015 4:14 PM **To:** Julian Morse <jmorse@mvlwb.com>

Cc: Tyree Mullaney <tyree@mvlwb.com>; Jen Potten <jpotten@mvlwb.com>

Subject: NATCL - Comment Response Extension Request

Importance: High

Hello Julian,

Given the very late submission of the DFN comments, NATCL would like to take an extra day to review and response to these comments.

We are requesting an extension of the deadline to Friday, October 16th.

Thank you

Deborah Flemming, Environmental Superintendent North American Tungsten Corp. Cantung Minesite PO Box 848 Watson Lake, Yukon YOA 1CO Ph. <u>604 759 0913</u> L275 FAX <u>604 759 0918</u>

Email: dflemming@natcl.ca

Cantung Enviro: cantungenviro@natcl.ca
Mactung Project: mactung@natcl.ca

"Safety is not expensive - It's priceless"

This e-mail may contain privileged and confidential material and its transmission is not a waiver of that privilege. It is intended for the sole use of the person to whom it is addressed. Any copying, disclosure, distribution or reliance on this material by anyone other than the intended recipient is strictly prohibited. We assume no responsibility to persons other than the intended recipient. If you have received this transmission in error, please notify the sender immediately and destroy any hard copies you may have printed and remove all copies of the email from your hard drive.